Author's response to reviews

Title: Comparison of clinical course and outcome of acute pancreatitis according to the two main etiologies: alcohol and gallstone

Authors:

Joon Hyun Cho (ygowgo96@hanmail.net)
Tae Nyeun Kim (tnkim@yu.ac.kr)
Sung Bum Kim (jeje0620@naver.com)

Version: 4 Date: 11 June 2015

Author's response to reviews:

Dear editorial board of BMC Gastroenterology,
Thank you very much for your kind comments.

1) As described in the “Acknowledgements”, our manuscript was copyedited by a professional language editing service for English grammar and style.
2) We described in the “Methods section” whether written informed consent for participation in the study was obtained.